\begin{table}%t3
\caption{Observational values of fractional abundances in L134N, and corresponding values at the "best fit'' times for each model. }
\tiny
\label{tab3}
\small
%\centerline
{\begin{tabular}{lccccc}
\hline
\hline
\noalign{\smallskip}
 Species & $N(i) / N($H$_{2})$$^{1,2}$ & \multicolumn{4}{c}{$n(i) / n($H$_{2})$ $^{1,3}$} \\
%\noalign{\smallskip}
 & Observed & M0 & M1 & M2 & M3 \\
%\noalign{\smallskip}
 & (L134N) & $t=2.6$ Myr & $t=1.9$ Myr & $t=2.8$ Myr & $t=3.1$ Myr \\
%\noalign{\smallskip}
\hline
\noalign{\smallskip}
C            & $\geq$1.0(--6)$^{4}$ &  6.2(--7) & {\bfseries  8.1(--8)} &  5.3(--7) &  2.0(--7) \\
%\noalign{\smallskip}
CH           & 1.0(--8)  $^{5}$ &  6.8(--9) &  3.2(--9) &  7.7(--9) &  7.7(--9) \\
%\noalign{\smallskip}
C$_2$H       & $\leq$5.0(--8)$^{5}$ &  8.5(--9) &  5.5(--9) &  9.3(--9) &  1.1(--8) \\
%\noalign{\smallskip}
C$_3$H       & 3.0(--10)  $^{5}$ & {\bfseries \em  6.8(--9)} &  1.5(--9) & {\bfseries \em  8.6(--9)} & {\bfseries \em  6.9(--9)} \\
%\noalign{\smallskip}
C$_3$H$_2$   & 2.0(--9)  $^{5}$ & {\bfseries  1.0(--10)} & {\bfseries  3.6(--11)} &  2.2(--10) &  3.9(--10) \\
%\noalign{\smallskip}
C$_3$H$_4$   & $\leq$1.2(--9)  $^{5}$ &  4.0(--11) &  2.4(--10) &  3.5(--10) &  1.4(--9) \\
%\noalign{\smallskip}
C$_4$H       & 1.0(--9)  $^{5}$ &  3.9(--9) &  1.1(--9) &  4.7(--9) &  5.3(--9) \\
%\noalign{\smallskip}
CN           & 8.2(--10)  $^{5}$ & {\bfseries \em  3.3(--8)} & {\bfseries \em  1.4(--8)} & {\bfseries \em  4.3(--8)} & {\bfseries \em  4.8(--8)} \\
%\noalign{\smallskip}
HCN          & 1.2(--8)  $^{6}$ &  4.5(--8) &  2.0(--8) &  6.9(--8) &  8.3(--8) \\
%\noalign{\smallskip}
HNC          & 4.7(--8)$^{6}$ &  4.0(--8) &  1.9(--8) &  6.3(--8) &  7.5(--8) \\
%\noalign{\smallskip}
H$_2$CN$^+$  & $\leq$3.1(--9)$^{5}$ &  8.3(--10) &  3.6(--10) &  1.1(--9) &  1.2(--9) \\
%\noalign{\smallskip}
CH$_2$CN     & $\leq$1.0(--9)$^{5}$ &  3.3(--10) &  1.0(--10) &  6.1(--10) &  8.8(--10) \\
%\noalign{\smallskip}
CH$_3$CN     & $\leq$1.0(--9)$^{5}$ &  1.1(--10) &  2.3(--11) &  2.0(--10) &  2.4(--10) \\
%\noalign{\smallskip}
C$_3$N       & $\leq$2.0(--10)$^{5}$ &  4.3(--10) &  1.0(--10) &  5.5(--10) &  5.5(--10) \\
%\noalign{\smallskip}
C$_3$H$_3$N  & $\leq$1.0(--10)$^{5}$ &  1.6(--12) &  7.7(--13) &  5.7(--12) &  2.3(--11) \\
%\noalign{\smallskip}
HC$_3$N      & 8.7(--10)$^{6}$ &  2.2(--10) &  1.1(--10) &  4.3(--10) &  8.5(--10) \\
%\noalign{\smallskip}
HC$_5$N      & 1.0(--10)$^{5}$ &  2.2(--11) & {\bfseries  7.5(--12)} &  4.6(--11) &  1.2(--10) \\
%\noalign{\smallskip}
HC$_7$N      & 2.0(--11)$^{5}$ & {\bfseries  5.9(--13)} & {\bfseries  1.4(--13)} & {\bfseries  1.6(--12)} &  5.1(--12) \\
%\noalign{\smallskip}
CO           & 8.0(--5)$^{5}$ &  1.2(--5) &  2.2(--5) &  1.4(--5) &  2.3(--5) \\
%\noalign{\smallskip}
HCO$^+$      & 1.0(--8)$^{6}$ &  2.1(--9) &  3.0(--9) &  2.3(--9) &  3.0(--9) \\
%\noalign{\smallskip}
H$_2$CO      & 2.0(--8)$^{5}$ &  2.1(--8) &  1.3(--8) &  3.7(--8) &  7.8(--8) \\
%\noalign{\smallskip}
CH$_3$OH     & 3.7(--9)$^{6}$ & {\bfseries  6.0(--13)} &  1.7(--9) &  4.1(--9) &  1.9(--8) \\
%\noalign{\smallskip}
HCOOH        & 3.0(--10)$^{5}$ & {\bfseries  1.5(--11)} &  1.5(--10) &  3.0(--11) &  1.8(--10) \\
%\noalign{\smallskip}
CH$_2$CO     & $\leq$7.0(--10)$^{5}$ &  2.9(--10) &  3.9(--10) &  3.2(--10) &  4.1(--10) \\
%\noalign{\smallskip}
CH$_3$CHO    & 6.0(--10)$^{5}$ & {\bfseries  1.3(--13)} & {\bfseries  5.6(--13)} & {\bfseries  1.3(--12)} & {\bfseries  7.0(--12)} \\
%\noalign{\smallskip}
C$_3$O       & $\leq$5.0(--11)$^{5}$ &  1.0(--11) &  7.2(--12) &  1.4(--11) &  2.1(--11) \\
%\noalign{\smallskip}
H$_2$S       & 8.0(--10)$^{5}$ & {\bfseries  1.6(--11)} &  2.9(--9) &  3.1(--9) &  3.1(--9) \\
%\noalign{\smallskip}
SO           & 3.1(--9)$^{6}$ & {\bfseries  6.9(--11)} &  3.2(--9) &  1.5(--9) &  4.4(--9) \\
%\noalign{\smallskip}
SO$_2$       & $\leq$1.6(--9)$^{6}$ &  8.9(--13) &  7.6(--11) &  3.2(--11) &  3.3(--10) \\
%\noalign{\smallskip}
CS           & 1.7(--9)$^{6}$ &  6.8(--10) &  6.1(--9) &  1.3(--8) & {\bfseries \em  1.9(--8)} \\
%\noalign{\smallskip}
HCS$^+$      & 6.0(--11)$^{5}$ & {\bfseries  2.7(--12)} &  2.1(--11) &  5.0(--11) &  6.3(--11) \\
%\noalign{\smallskip}
H$_2$CS      & 6.0(--10)$^{5}$ &  2.0(--10) &  1.1(--9) &  5.1(--9) & {\bfseries \em  8.4(--9)} \\
%\noalign{\smallskip}
C$_2$S       & 6.0(--10)$^{5}$ & {\bfseries  2.5(--11)} &  1.5(--10) &  3.5(--10) &  2.6(--10) \\
%\noalign{\smallskip}
C$_3$S       & $\leq$2.0(--10)$^{5}$ &  3.7(--12) &  1.4(--11) &  5.6(--11) &  4.6(--11) \\
%\noalign{\smallskip}
OCS          & 2.0(--9)$^{5}$ & {\bfseries  4.0(--12)} & {\bfseries  6.2(--11)} & {\bfseries  7.5(--11)} & {\bfseries  1.4(--10)} \\
%\noalign{\smallskip}
NH$_3$       & 9.1(--8)$^{5}$ &  1.7(--7) &  1.2(--7) &  2.3(--7) &  2.7(--7) \\
%\noalign{\smallskip}
N$_2$H$^+$   & 6.8(--10)$^{6}$ &  2.3(--9) &  2.1(--9) &  2.5(--9) &  2.2(--9) \\
%\noalign{\smallskip}
NO           & 6.0(--8)$^{5}$ &  4.4(--8) &  9.1(--8) &  5.9(--8) &  1.4(--7) \\
%\noalign{\smallskip}
OH           & 7.5(--8)$^{5}$ &  3.6(--8) &  4.9(--8) &  4.9(--8) &  1.1(--7) \\
%\noalign{\smallskip}
H$_2$O       & $\leq$3.0(--7)$^{7}$ &  4.5(--8) &  8.6(--8) &  9.1(--8) &  2.6(--7) \\
%\noalign{\smallskip}
O$_2$        & $\leq$1.7(--7)$^{8}$ &  1.8(--7) & {\bfseries \em  2.6(--6)} &  2.0(--7) &  8.8(--7) \\
%\noalign{\smallskip}
\hline
%\noalign{\smallskip}
Matches$^{9}$ /41 && 29 & 33 & 36 & 35 \\
%\noalign{\smallskip}
\hline
\end{tabular}}
\medskip
{$^{1}$ $a(b) = a \times 10^{b}$; $^{2}$ Observed values collated by \cite{wakelam06b}; $^{3}$ Boldface indicates a }
{theoretical value different by more than 1 order of magnitude from the observed value. Plain} 
{boldface indicates too low a value; italic boldface indicates too high a value; $^{4}$ \cite{stark96a}; }\protect 
{$^{5}$ \cite{ohishi92a}; $^{6}$ \cite{dickens00a}; $^{7}$ \cite{snell00a}; $^{8}$ \cite{pagani03a}} {$^{9}$ Agreement with observational value, to within 1 order of magnitude.} 
\end{table}